Author’s response to reviews

Title: Childhood sensorineural hearing loss and adult mental health up to 43 years later: Results from the HUNT study

Authors:

Mariann Idstad (Mariann.Idstad@fhi.no)
Kristian Tambs (kristian.tambs@fhi.no)
Lisa Aarhus (Lisa.Aarhus@stami.no)
Bo Engdahl (BoLars.Engdahl@fhi.no)

Version: 2 Date: 09 Jan 2019

Author’s response to reviews:

Response to editor.

Editor Comments:

1. In Discussion you should maybe note (end of page 12) that some deaf people see themselves as part of a deaf minority culture (including use of sign language) and that research has shown that this group might be less likely to report low subjective well-being.

This has now been included at the end of page 12.

2. Reference 23 and 27 are the same.

This mistake has now been corrected: the reference number “27” in the text has been replaced with “23” and reference number 27 has been deleted from the reference list.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

All documents uploaded are final, clean versions.